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Program



Summary

• Beam setup on Monday (S. Bettoni, F. Loehl. C. Kittel)
Photon energy changed from 7.5 keV to 12.05 keV
Maintained short pulse duration (< 30 fs rms)

• 50 Hz operation during the entire week
• No major issues during the entire week
• Timing jitter between FEL pulses and pump-laser most of the time very good 

(down to 19 fs rms for 500 – 1000 shots – new record at Alvra!)

Smaller issues
• BAM observed large timing jitter at some periods of time

à Possibly issues with OMO1 or OMO2, will be investigated during the shutdown
• Did not run FEL pointing feedback due to problems with gas detector
• Undulator movers not available due to PLC problems

Achieved ~ 500 µJ of pulse energy most of the week.



Photon pulse energy during the week
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Undulator contributions

We always had problems with 
the first undulator segments 
not contributing to the lasing.

à Tried optimizing the 
undulator kick corrections with
PSICO.



Undulator contributions

Starting point: ~ 500 µJ

First segments do now contribute!
(Not all segments by the same amount 
yet) 

Thanks to E. Ferrari and E. Prat for running the script!


